Author’s response to reviews

Title: Association of environmental and sociodemographic factors with life satisfaction in 27 European countries

Authors:

Nikita Rajani (nikita.rajani14@imperial.ac.uk)
Vassilis Skianis (bill.skianis@gmail.com)
Filippos Filippidis (f.filippidis@imperial.ac.uk)

Version: 2 Date: 02 Apr 2019

Author’s response to reviews:

Technical Comments:

- We note that figure is not cited, please cite this in the main manuscript.

Response: We have cited Figure 1 in the Results section of the manuscript

Assistant Editor Comments:

1. Ethics Approval and Consent

-- Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate or reference relevant legislation if formal ethics approval was not required.

-- If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Response: We have added that the study was deemed exempt from ethical approval by the ethics committee of Imperial College.

2. Abstract

Please include the rationale of study in Background section of the abstract.

Response: We have edited the background section to include the rationale of the study.
3. Availability of Data and Materials

Please state whether the data was publically available or whether any permissions were required to access it.

Response: We have edited this statement to include that data were freely available from the website mentioned.

4. Headings

Please provide a figure legends heading.

Response: We have added a heading.

5. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: We submit a clean version of the manuscript.

----------------------Editorial Policies----------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

Response: We have made sure that all the above sections are included in the Declarations.